Infection after total joint replacement of the hip and knee: research programme including the INFORM RCT

Ashley W Blom,^{1,2*} Andrew D Beswick,² Amanda Burston,² Fran E Carroll,³ Kirsty Garfield,^{3,4} Rachael Gooberman-Hill,² Shaun Harris,^{3,4} Setor K Kunutsor,^{1,2} Athene Lane,^{3,4} Erik Lenguerrand,² Alasdair MacGowan,⁵ Charlotte Mallon,² Andrew J Moore,² Sian Noble,³ Cecily K Palmer,^{3,6} Ola Rolfson,^{2,7} Simon Strange² and Michael R Whitehouse^{1,2} on behalf of the INFORM group

Declared competing interests of authors: Ashley W Blom is a co-applicant on a grant from Stryker Corporation (Kalamazoo, MI, USA) for investigating the outcome of the Triathlon total knee replacement. He is a member of the National Joint Registry lot 2 contract (statistical analysis) team and a member of the National Institute for Health and Care Research (NIHR) Bristol Biomedical Research Centre. Erik Lenguerrand is also a member of the National Joint Registry lot 2 contract (statistical analysis) team. Ola Rolfson reports grants from Stryker Corporation, Pfizer Inc. (New York, NY, USA) and LINK Sweden AB (Åkersberga, Sweden) outside the submitted work. Michael R Whitehouse undertakes teaching on basic sciences for orthopaedic trainees preparing for the Fellowship of the Royal College of Surgeons; his institution receives market-rate payment for this teaching from Heraeus (Hanau, Germany). He undertakes teaching on total hip replacement for orthopaedic consultants and trainees;

¹National Institute for Health and Care Research Bristol Biomedical Research Centre, University Hospitals Bristol and Weston NHS Foundation Trust and University of Bristol, Bristol, UK

²Musculoskeletal Research Unit, Translational Health Sciences, Bristol Medical School, University of Bristol, Bristol, UK

³Population Health Sciences, Bristol Medical School, University of Bristol, Bristol, UK

⁴Bristol Randomised Trials Collaboration, University of Bristol, Bristol UK

⁵Department of Medical Microbiology, Southmead Hospital, Bristol, UK

⁶National Institute for Health and Care Research Applied Research Collaboration West (NIHR ARC West), University Hospitals Bristol NHS Foundation Trust and University of Bristol, Bristol, UK

⁷Department of Orthopaedics, Institute of Clinical Sciences, Sahlgrenska Academy, University of Gothenburg, Gothenburg, Sweden

^{*}Corresponding author Ashley.Blom@bristol.ac.uk

his institution receives market-rate payment for this teaching from DePuy Synthes (Raynham, MA, USA). He is a co-applicant on a grant from Stryker Corporation investigating the outcome of the Triathlon total knee replacement. He is a member of the National Joint Registry lot 2 contract (statistical analysis) team. He is NIHR Clinical Research Network West of England lead for Trauma and Emergencies, a member of the NIHR Bristol Biomedical Research Centre and principal investigator on two NIHR Health Technology Assessment grants (NIHR129011 and NIHR127849).

Published November 2022 DOI: 10.3310/HDWL9760

Plain English summary

Research programme including the INFORM RCT

Programme Grants for Applied Research 2022; Vol. 10: No. 10

DOI: 10.3310/HDWL9760

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

People with severe osteoarthritis, other joint conditions or joint injuries may undergo joint replacement to reduce pain and disability. In the UK in 2019, over 200,000 hip and knee replacements were performed. About 1 in 100 become infected. Treatment usually requires two operations to remove and replace the joint, with antibiotics between surgeries. Some surgeons treat joint infection with one operation.

Our research was needed to find out why some patients are predisposed to getting joint infections and how this affects patients and the NHS, and to evaluate treatments.

We reviewed previous research, analysed a national joint registry, interviewed patients and surgeons to find out their experiences of infection, assessed costs to the NHS and patients, and explored aspects of treatments important to patients. Treatments were compared by randomly allocating 140 patients with hip joint infection to one or two operations and assessing the impact on quality of life and health-care costs. A patient forum supported the research.

We found that, after hip and knee replacement, about 0.62% and 0.75% of patients, respectively, had joint infection requiring surgery. It costs over £30,000 to treat a hip joint infection.

We showed that risk of joint infection is greater in men, people who are overweight and those with pre-existing health conditions, and when some surgical techniques are used. Joint infection is difficult to detect, but new tests of joint fluid show promise. Patients and surgeons described the devastating effects of joint infection. Important concerns for patients were the time taken to recover and engage in valued activities and the need for support and information.

The research we reviewed indicated that hip joint infection treated in one or two operations cleared infection equally, but joint registry analysis raised concern about early problems after treatment with one operation. The randomised trial found that recovery was delayed in people receiving two operations. However, after 18 months, the levels of pain, disability and complications were similar between the groups. The NHS and patient costs were lower when treatment was with one operation.

Programme Grants for Applied Research

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

Programme Grants for Applied Research (PGfAR) was launched in 2013 and is indexed by Europe PMC, NCBI Bookshelf, DOAJ, Ulrichsweb™ (ProQuest LLC, Ann Arbor, MI, USA) and Scopus® (Elsevier, Amsterdam, Netherlands).

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar.

Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Programme Grants for Applied Research programme

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health and Care Research (NIHR), was established in 2006 to fund collaborative, multidisciplinary programmes of applied research to solve health and social care challenges. Findings are expected to provide evidence that lead to clear and identifiable patient benefits, in the relatively near future.

PGfAR is researcher led and does not specify topics for research; however, the research must be in an area of priority or need for the NHS and the social care sector of the Department of Health and Social Care, with particular emphasis on health and social care areas that cause significant burden, where other research funders may not be focused, or where insufficient funding is available.

The programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director. For more information about the PGfAR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/programme-grants-for-applied-research.htm

This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-1210-12005. The contractual start date was in February 2014. The final report began editorial review in June 2020 and was accepted for publication in May 2021. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, PGfAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the PGfAR programme or the Department of Health and Social Care.

Copyright © 2022 Blom et al. This work was produced by Blom et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Dr Cat Chatfield Director of Health Services Research UK

NIHR Journals Library Editors

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editor-in-Chief of HSDR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Dr Peter Davidson Interim Chair of HTA and EME Editorial Board. Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Rob Riemsma Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk